Author's response to reviews

Title: A systematic review including meta analysis of work environment and depressive symptoms Running title: Job strain and bullying at work are related to depressive symptoms

Töres Theorell, PhD, MD, Stress Research Institute, Stockholm University, SE-106 91, Stockholm, Sweden, and Department of Neuroscience, Karolinska Institutet, SE-171 77 Stockholm, Sweden, tores.theorell@stressforskning.su.se Gunnar Aronsson, PhD, Department of Psychology, Stockholm University, SE-106 91, Stockholm, Sweden, gunnar.aronsson@psychology.su.se Lil Träskman Bendz, PhD, MD Division of Psychiatry, Lund University, SE-221 00 Lund, Sweden, lil.traskman_bendz@med.lu.se Tom Grape, MD, South Health Care Centre, SE-761 46, Norrtälje, Sweden, tom.grape@tiohundra.se Anne Hammarström, PhD, MD, Department of Public Health and Clinical Medicine, Department of social medicine, Umea University Hospital, SE-901 85 Umea, Sweden, anne.hammarstrom@fammed.umu.se Christer Hogstedt, PhD, MD, Division of Occupational medicine

Authors:

Töres Theorell (Tores.Theorell@stressforskning.su.se)
Gunnar Aronsson (Gunnar.Aronsson@psychology.su.se)
Lil Träskman Bendz (lil.traskman_bendz@med.lu.se)
Tom Grape (tom.grape@tiohundra.se)
Anne Hammarström (anne.hammarstrom@fammed.umu.se)
Christer Hogstedt (christer.hogstedt@gmail.com)
Ina Marteinsdottir (ina.marteinsdottir@liu.se)
Ingmar Skoog (ingmar.skoog@neuro.gu.se)
Charlotte Hall (Charlotte.Hall@sbu.se)

Version: 2 Date: 9 March 2015

Author's response to reviews:

We have followed the instructions from the editorial office